Author's response to reviews

Title: Gene sequence variations of the platelet P2Y12 receptor are associated with coronary artery disease

Authors:

Ugo Cavallari (ugo.cav@fastwebnet.it)
Elisabetta Trabetti (elisabetta.trabetti@univr.it)
Giovanni Malerba (giovanni.malerba@univr.it)
Michele Biscuola (michele.biscuola@gmail.com)
Domenico Girelli (domenico.girelli@univr.it)
Oliviero Olivieri (oliviero.olivieri@univr.it)
Nicola Martinelli (trentinik@jumpy.it)
Dominick J Angiolillo (dominick.angiolillo@jax.ufl.edu)
Roberto Corrocher (roberto.corrocher@univr.it)
Pier Franco Pignatti (pierfranco.pignatti@univr.it)

Version: 4 Date: 4 September 2007

Author's response to reviews: see over
Dear BioMed Central Editorial Production Team,

we made the formatting changes, as requested, in the revised version of the manuscript entitled “Gene sequence variations of the platelet P2Y\textsubscript{12} receptor are associated with coronary artery disease” by U. Cavallari et al.

Kind regards,
Yours sincerely

Elisabetta Trabetti

Address for correspondence:
Elisabetta Trabetti,
Department of Mother and Child and Biology-Genetics, Section of Biology-Genetics. University of Verona.
Strada Le Grazie 8, 37134 Verona, Italy.
Phone: +39-45-584602  Fax: +39-45-8027180  E-mail: elisabetta.trabetti@univr.it